Author's response to reviews

Title: Underestimation of weight and its associated factors among overweight and obese adults in Pakistan: a cross sectional study

Authors:

Seema Bhanji (seema.bhanji@aku.edu)
Ali Khan Khuwaja (ali.khuwaja@aku.edu)
Fawad Siddiqui (fawadsiddiqui@hotmail.com)
Iqbal Azam (iqbal.azam@aku.edu)
Khawar Kazmi (khawar.kazmi@aku.edu)

Version: 12 Date: 18 May 2011

Author's response to reviews:

May 18, 2011,

Editor,
BMC Public Health

Dear Sir,

Kindly find attached the revised manuscript, which has been proof read for typographical and content errors.

Thanks and Best Regards,

Dr. Seema Bhanji
Corresponding author